Author's response to reviews

Title: Baseline assessment of adult and adolescent primary care delivery in Rwanda: an opportunity for quality improvement

Authors:

Ashwin Vasan (ashwin.vasan@lshtm.ac.uk)
Manzi Anatole (mangano2020@gmail.com)
Catherine Mezzacappa (catherine.mezzacappa@gmail.com)
Bethany L Hedt-Gauthier (bethhedt@gmail.com)
Lisa R Hirschhorn (lrh3@hms.harvard.edu)
Fulgence Nkikahizi (fuldgence@gmail.com)
Marc Hagenimana (hagmarc@gmail.com)
Aphrodis Ndayisaba (aphronda@gmail.com)
Felix R Cyamatare (cyaferwabu@gmail.com)
Bonaventure Nzeyimana (nzeyimana.bonaventure68@gmail.com)
Peter Drobac (pdrobac@pih.org)
Neil Gupta (ngupta5@partners.org)

Version: 2 Date: 15 August 2013

Author's response to reviews:

Dear Sir(s), Mdm(s):

Thank you for your consideration of our manuscript: Baseline assessment of adult and adolescent primary care delivery in Rwanda: an opportunity for quality improvement. As advised in your initial response, we have made the following revisions to the manuscript:

1. A statement on informed consent has been included in the "Ethics" sub-section within the Methods.
2. A statement on competing interests has been drafted and included after Conclusions
3. A section on Authors' Contributions has been included after the Competing Interests section.

We hope that you find these revisions satisfactory and look forward to your continued review and consideration of our manuscript. We look forward to your response and hope for a positive outcome.

Sincerely,

Ashwin Vasan, M.D., M.Sc. (on behalf of the Authors)